\magnification = \magstep 1
\vsize 9 true in
\hsize 6.5 true in
\def\Z{{\Bbb Z}}
\def\({\left(}
\def\){\right)}
\def\goodbreak{\par\penalty-500 }
\def\problem#1\par{\bigskip\bigskip\goodbreak\centerline{Problem #1}\bigskip}
\define\Aut{\operatorname{Aut}}
\define\End{\operatorname{End}}
\font\bigbf=cmbx12
\def\ni{\noindent}
\def\bs{\bigskip}
\def\ms{\medskip}
\def\ss{\smallskip}
\def\subhead{\bf}
\centerline{ \bigbf Homework, October 16, 1989 }
\bs
\centerline { \bigbf Math 401 -- Second Course in Abstract Algebra, I}
\ms
\centerline { \bigbf Prof. Daniel Grayson }
\ms
\centerline { 2:00 PM MWF, Section F1, 241 Altgeld Hall }
\bs

{\bf Homework:} Read VII.1 and do exercises 2, 6, 8.  Read VII.2 and its
appendix and do exercises 7 and 12.

\end